Author’s response to reviews

Title: Multi-cohort modeling strategies for scalable globally accessible prostate cancer risk tools

Authors:
Johanna Tolksdorf (johanna.tolksdorf@tum.de)
Michael Kattan (kattanm@ccf.org)
Stephen Boorjian (boorjian.stephen@mayo.edu)
Stephen Freedland (Stephen.freedland@cshs.org)
Karim Saba (Karim.saba@usz.ch)
Cedric Poyet (Cedric.Poyet@usz.ch)
Lourdes Guerrios (Lourdes.guerrios1@upr.edu)
Amanda De Hoedt (Amanda.Dehoedt@va.gov)
Michael Liss (liss@uthscsa.edu)
Robin Leach (leach@uthscsa.edu)
Javier Hernandez (hernandezJ7@uthscsa.edu)
Emily Vertosick (vertosie@mskcc.org)
Andrew Vickers (vickersa@mskcc.org)
Donna Ankerst (ankerst@tum.de)

Version: 3 Date: 11 Sep 2019

Author’s response to reviews:

1. Ethics Approval

   -- Please include the full name of all the ethics committees in each country (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate under 'ethics approval and consent to partipate'. If there are too many to list this can be uploaded as an additional file and cited under ethics approval.
2. Consent

-- Please clarify whether informed consent was obtained from the participants. If consent to participate was not required, please provide details of why this was waived and the institution/committee that approved the waiver.

SENTENCE ADDED

3. Authors Contributions

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

DONE

4. Acknowledgements

If there are no acknowledgements, please state “Not applicable” under this heading.

DONE

5. Formatting

-- Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

CONFIRMED